Author's response to reviews

Title: The Professional Nurse Self-Assessment Scale: psychometric testing in Norwegian long term and home care contexts

Authors:

elisabeth finnbakk (elisabeth.finnbakk@ldh.no)
sigrid wangensteen (sigrid.wangensteen@hig.no)
kirsti skovdahl (kirsti.skovdahl@hbv.no)
lisbeth fagerstrom (lisbeth.fagerstrom@hbv.no)

Version: 6 Date: 21 October 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for your acceptance of our manuscript for publication in your journal. As per your latest requirement, the title page is included at the beginning of the manuscript – and removed from the additional files. Please see page 1 and 2 in the main manuscript.

We, Sigrid Wangensteen (SW), Kirsti Skovdahl (KS), Lisbeth Fagerström (LF), and myself (Elisabeth Finnbakk; EF), have all been involved in the completion of the manuscript. The authorship contributions are as follows: study design EF, SW, LF; instrument development EF, SW, KS, LF; data collection EF; analysis EF, SW, LF; and manuscript preparation EF, SW, KS, LF. The manuscript has not been previously published. All authors meet the criteria for authorship and have approved the final manuscript. Lastly, we the authors declare no competing interests.

In regard to publication fees, the corresponding author, Elisabeth Finnbakk, can organize payment by credit card as necessary. We are extremely pleased to have been accepted for publication, and thank you once again for all of your help and interest during the approval process.

Yours sincerely,

Elisabeth Finnbakk, Corresponding author
Lovisenberg Diaconal University College
Lovisenbergt. 15 b
0456 Oslo
Norway

E-mail: elisabeth.finnbakk@ldh.no
Mobile phone: +47 975 444 09